Author's response to reviews

Title: Quality of Care Associated with Number of Cases Seen and Self-reports of Clinical Competence for Japanese Physicians-in-training in Internal Medicine

Authors:

Yasuaki Hayashino (hayasino@kuhp.kyoto-u.ac.jp)
Shunich Fukuhara (fukuhara@pbh.med.kyoto-u.ac.jp)
Kunihiko Matsui (kmatsui@kaiju.medic.kumamoto-u.ac.jp)
Yoshinori Noguchi (yoshi-noguchi@umin.ac.jp)
Taro Minami (nantaro@gmail.com)
Dan Bertenthal (dberten@itsa.ucsf.edu)
John W Peabody (peabody@psg.ucsf.edu)
Yoshitomo Mutoh (ym6104@toranomon.gr.jp)
Yoshihiko Hirao (hirao-yos@naramed-u.ac.jp)
Kazuhiko Kikawa (kkkikawa@fc.kuh.kumamoto-u.ac.jp)
Yohei Fukumoto (yoheifuk@yamaguchi-u.ac.jp)
Junichiro Hayano (hayano@med.nagoya-cu.ac.jp)
Teruo Ino (tino@fujita-hu.ac.jp)
Umihiko Sawada (usawada@med.nihon-u.ac.jp)
Jin Seino (S7919@snh.go.jp)
Norio Higuma (nhiguma@hosp.niigata.niigata.jp)
Hiroyasu Ishimaru (ishi-hir@mx5.nisiq.net)

Version: 5 Date: 12 June 2006

Author's response to reviews:

Dear Editors,

Electrically submitted is our revised manuscript "Quality of Care Associated with Number of Cases Seen and Self-reports of Clinical Competence for Japanese Physicians-in-training in Internal Medicine" We revised our manuscript considering the formatting changes requested.

Thank you.

Sincerely yours

Yasuaki Hayashino, MD, MPH
Department of Epidemiology and Healthcare Research Kyoto University
Graduate School of Medicine Kone-cho, Japan
Yoshida, Sakyoku, Kyoto 606-8501
Phone 075-753-9467
FAX 075-753-4644
hayasino-y@umin.net